

GOPEN ACCESS

Citation: McDonald SA, Jan van Hoek A, Paolotti D, Hooiveld M, Meijer A, de Lange M, et al. (2024) A statistical modelling approach for determining the cause of reported respiratory syndromes from internet-based participatory surveillance when influenza virus and SARS-CoV-2 are co-circulating. PLOS Digit Health 3(12): e0000655. https://doi.org/ 10.1371/journal.pdig.0000655

Editor: Crina Grosan, King's College London, UNITED KINGDOM OF GREAT BRITAIN AND NORTHERN IRELAND

Received: July 26, 2023

Accepted: September 27, 2024

Published: December 9, 2024

Copyright: © 2024 McDonald et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The weekly ARI numbers from Infectieradar, the virological testing data used in this research, and the JAGS code used for modelling attribution are included in the Supporting Information.

Funding: This work was supported by the European Union's Horizon Europe VERDI project (Grant agreement number 101045989) to DP. This RESEARCH ARTICLE

A statistical modelling approach for determining the cause of reported respiratory syndromes from internet-based participatory surveillance when influenza virus and SARS-CoV-2 are co-circulating

Scott A. McDonald¹, Albert Jan van Hoek¹, Daniela Paolotti^{2*}, Mariette Hooiveld³, Adam Meijer¹, Marit de Lange¹, Infectieradar team¹, Arianne van Gageldonk-Lafeber¹, Jacco Wallinga^{1,4}

1 Centre for Infectious Disease Control, Netherlands National Institute for Public Health and the Environment, Bilthoven, the Netherlands, 2 Institute for Scientific Interchange Foundation, Torino, Italy, 3 Nivel, Utrecht, the Netherlands, 4 Department of Biomedical Data Sciences, Leiden University Medical Center, Leiden, the Netherlands

¶ Membership of the Infectieradar team is provided in the Acknowledgement * daniela.paolotti@isi.it

Abstract

Symptom-only case definitions are insufficient to discriminate COVID-like illness from acute respiratory infection (ARI) or influenza-like illness (ILI), due to the overlap in case definitions. Our objective was to develop a statistical method that does not rely on case definitions to determine the contribution of influenza virus and SARS-CoV-2 to the ARI burden during periods when both viruses are circulating. Data sources used for testing the approach were weekly ARI syndrome reports from the Infectieradar participatory syndromic surveillance system during the analysis period (the first 25 weeks of 2022, in which SARS-CoV-2 and influenza virus co-circulated in the Netherlands) and data from virologically tested ARI (including ILI) patients who consulted a general practitioner in the same period. Estimation of the proportions of ARI attributable to influenza virus, SARS-CoV-2, or another cause was framed as an inference problem, through which all data sources are combined within a Bayesian framework to infer the weekly numbers of ARI reports attributable to each cause. Posterior distributions for the attribution proportions were obtained using Markov Chain Monte-Carlo methods. Application of the approach to the example data sources indicated that, of the total ARI reports (total of 11,312; weekly mean of 452) during the analysis period, the model attributed 35.4% (95% Crl: 29.2–40.0%) and 27.0% (95% Crl: 19.3–35.2%) to influenza virus and SARS-CoV-2, respectively. The proposed statistical model allows the attribution of respiratory syndrome reports from participatory surveillance to either influenza virus or SARS-CoV-2 infection in periods when both viruses are circulating, but comparability of the participatory surveillance and virologically tested populations is important. Portability for use by other countries with established participatory respiratory surveillance systems is an asset.

work reflects only the authors' views. The European Commission is not responsible for any use that may be made of the information it contains. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

Internet-based participatory surveillance systems are valuable public health tools for monitoring illness due to infectious diseases, but when two respiratory viruses are circulating simultaneously, but lead to similar symptoms, case definitions cannot be used to distinguish the virus causing the illness. Our goal was to develop a statistical method that does not rely on case definitions to determine the contribution of influenza virus and SARS-CoV-2 to reported cases of acute respiratory infection. We tested the method using data collected from the Infectieradar system during the 2021/2022 influenza season in the Netherlands, in a period in which the omicron variant of SARS-CoV-2 was also circulating. Our study indicates that the model is able to attribute illness to the possible causes, and is of potential value for other countries with established participatory respiratory surveillance systems.

Introduction

Internet-based participatory syndromic surveillance systems have proven value for providing (near) real-time information on the spread of seasonal influenza in the general population [1,2,3], through voluntary self-reporting of respiratory symptoms that are consistent with influenza-like illness (ILI). Trends in self-reported ILI have been demonstrated to correlate well with trends in influenza virus infection as determined using other surveillance systems [1,2,4].

As COVID-19 appears to be on the path towards becoming an established recurring infectious disease, temporal overlap with seasonal influenza virus is inevitable. To maximise their utility for tracking either infection, syndromic surveillance systems will ideally need to distinguish syndrome reports attributable to the two (or more) causal agents, especially as several countries intend these systems to serve in the future as a monitor of the community spread of SARS-CoV-2 infection in the absence of widespread testing and collection of incident case data. For instance, in the Netherlands as of 11 April 2022, undergoing a confirmation PCR test at a governmental facility was no longer advised following a positive self-test (rapid antigen test).

Symptom-only case definitions are insufficient to discriminate between ILI and COVIDlike illness, due to the overlap in symptoms comprising the two case definitions (S2 Fig). This issue is apparent for the Netherlands in 2022, where the influenza season [5] partially overlapped with a period in which the SARS-CoV-2 positivity rate from national surveillance also exhibited a peak [6], but with different temporal dynamics.

The main aim of this work is to propose a method using Bayesian statistical modelling to perform syndrome attribution, which we test by combining data from the Infectieradar participatory surveillance system in the Netherlands with data from virological testing of GP patients with acute respiratory infection (ARI). This is an inference problem, the goal of which is the estimation of the weekly numbers of ARI reports that are attributable to each cause. Carrying out this inference within a Bayesian framework provides the flexibility to integrate additional relevant data, such as the prevalence of symptom(s) highly specific for one of the causes. Like broadly similar statistical approaches developed to attribute mortality to influenza virus and other respiratory pathogens [7,8], ecological assumptions are required when the data indicating confirmed influenza virus and SARS-CoV-2 infection are not collected from the same participants who self-report respiratory symptom(s). The ultimate goal of developing this methodology is two-fold: first, we wish to better estimate the incidence of both infections (influenza virus and SARS-CoV-2) among persons with ARI in the community (not only among people who consult their GP). Second, to support this effort, we would like to establish an operational approach to provide a clear estimate of the number of virological tests that are required to attribute respiratory syndrome occurrence to the causal agent. The latter is especially important for assessing the resources needed in terms of laboratory testing to implement a cost-effective approach for syndrome attribution, while at the same time leveraging the scale-up capabilities of participatory surveillance.

Methods

We consider the task of estimating the proportions of respiratory syndromes that are due to influenza virus, SARS-CoV-2, or another pathogen/cause, as an attribution problem, which is solved within a Bayesian framework by essentially inferring the relationships between variation in influenza positives and SARS-CoV-2 positives over time from virological testing, and variation in the time-series of ARI syndrome reports from participatory surveillance. The analysis period was defined as ISO weeks 1–25 of 2022, and was chosen to encompass the 2021/2022 influenza season, which lasted from week 8 through week 20 [5].

Data sources and case definitions

The method requires two main data sources: syndromic reports from participatory surveillance systems and a second source consisting of virological testing of swabbed patients for both influenza virus and SARS-CoV-2. Ideally the characteristics (e.g., demographic variables, symptom severity) of the tested population should match the characteristics of the participants self-reporting the syndrome of interest. This is not essential for presenting and developing the methodology, but will be important in practice. Virological testing of hospitalised patients could also serve as a viable data source, but then interpretation of the resulting attributions must take into account differences in the severity of illness and testing strategies.

Syndromic surveillance

The Infectieradar web-based syndromic surveillance system (participatory surveillance) was set up in mid-March 2020 and collates data on symptom occurrence, SARS-CoV-2 testing, vaccination against influenza virus and COVID-19, absenteeism, etc. through weekly surveys filled out by the respondents. As of June 2022, there were approximately 9000 active participants. From Infectieradar data, we calculated the number of submitted surveys per week that matched the case definition used by the European Centre for Disease Control [9] for acute respiratory infection. ARI is defined as at least one of the following symptoms: sore throat, cough, shortness of breath, rhinitis, with sudden onset of symptom(s) indicated [10]. ARI was chosen as syndrome case definition to match the study population of the virological testing data (described below). As in previous studies of involving Infectieradar and other national participatory surveillance systems belonging to the InfluenzaNet monitoring network [11], exclusion criteria were applied to reduce the chance of selection bias (i.e., persons who only registered and/or completed a single survey, likely because of prevalent symptoms) [4,12]. Namely, weekly surveys for which symptom onset date was prior to or on the date of registration were removed, and a participant's first-ever weekly survey was excluded. Note that a given participant can contribute data to multiple weeks of the analysis period if symptoms persist or reappear; thus we model *prevalent* rather than *incident* syndrome occurrence.

Loss of smell or taste has a high specificity for SARS-CoV-2 infection [13] but occurs infrequently in influenza cases [14]. We retrieved the weekly prevalence of this 'joint' symptom among Infectieradar ARI reports, to improve the positive predictive value of ARI for SARS--CoV-2 (see below).

Virological testing

The laboratory testing study population we used to develop and test the attribution model consisted of persons with ARI (consisting of primarily ILI patients, but also non-ILI ARI patients were swabbed) consulting a general practitioner (GP) belonging to a long-established sentinel network of GP practices (the Nivel Primary Care Database; [15]). ARI patients were judged as such by the GP, and were neither systematically nor randomly sampled. GPs were requested to sample at least two ARI patients per week, of which at least one with ILI and one younger than 10 years old. If no ILI patients presented, then ARI patients were sampled. Also, if patients had tested SARS-CoV-2 positive at home using a rapid antigen test they would normally not be coded as ARI but with a specific ICPC code for COVID-19 cases. Such patients would thus be excluded from sampling for virological testing. We note that this sampling protocol is adequate for purposes of developing the attribution method, but not for accurate attribution. Swabs (nose and throat swab) were tested for influenza A virus (with subtyping for A(H1N1) pdm09 and A(H3N2)), influenza B virus (with lineage determination for B/Victoria and B/ Yamagata), and SARS-CoV-2 at RIVM's Centre for Infectious Disease Research, Diagnostics and Laboratory Surveillance (IDS) (for further details on the virological testing procedure, see [16]). Samples are additionally tested for RSV, rhinovirus, enterovirus, other human coronaviruses (hCoV-229E, -OC43, -HKU1, -NL63), parainfluenza viruses (type 1, 2, 3, 4), and human metapneumovirus. Over our 25-week analysis period a total of 1074 samples were tested (mean of 43 samples per week; range 6 to 87).

The 2021/22 influenza season in the Netherlands began later than in a typical season, lasting from week 8 through week 20, with mainly influenza virus type A being detected, of which A (H3N2)

was dominant, with lower proportions of A(H1N1)pdm09 and influenza virus type B/Victoria lineage detected [5]. The peak in ILI incidence was reached approximately in week 15. The pattern for SARS-CoV-2 differed; from national surveillance of the weekly percent positive among tested persons, an earlier peak in week 10 (6.0% SARS-CoV-2 positive) was apparent [6],

Attribution model

The attribution model consisted of the following set of equations, in which index *i* to refers to week number of the analysis period, and the index *c* to refer to the (unknown) cause of the syndrome report.

We model the observed data, Z_i , which is a summation of the (unknown) number of ARI syndrome reports per cause (i.e., $Y_{influenza,i} + Y_{SARS-CoV-2,i} + Y_{other,i}$) using the *sum* sampler in JAGS [17]:

$$Z_i = \sum_c Y_{c,i} \tag{1}$$

The weekly number of syndrome reports attributable to three 'cause categories': influenza virus, SARS-CoV-2, and other (which, besides other respiratory pathogens, may include conditions such as allergies that can give rise to ARI symptoms), were each assumed to follow a Poisson distribution (appropriate for modelling event counts within a given time interval). $Y_{s,a,i}$ is the number of ARI syndrome reports in week *i* attributable to cause *c*.

$$Y_{c,i} \sim Poisson(\lambda_{c,i})$$
 (2)

The Poisson mean parameter, $\lambda_{c,i}$, is a function of the probability that the occurrence of the syndrome is attributable to cause c, $\pi_{c,i}$, and a scaling factor, δ_c , which can be interpreted as the expected number of ARI reports per unit probability of an ARI case being attributable to cause c. (Reliable estimation of δ_c requires only that the proportions positive for each cause category from virological testing do not closely co-vary, and that there is variation over time.)

$$\lambda_{c,i} = \delta_c \pi_{c,i} \tag{3}$$

Virological testing data (weekly proportion of samples positive for cause *c*, where *n* = number of positives; *N* = number of tested samples) are assumed binomially distributed (i.e., describing the number of outcomes out of *N* trials); these data inform the probability parameter $p_{c,b}$ for which a non-informative Jeffreys prior, Beta(0.5,0.5), is specified:

$$n_{c,i} \sim Binomial(N_{c,i}, p_{c,i})$$
 (4)

A vague half-normal prior is specified for the scaling factor δ_c (to ensure a positive density over the non-negative real numbers), with large variance:

$$\delta_c \sim half - Normal(0, 100000) \tag{5}$$

Next, we bring in further information on the weekly prevalence $(prev_i)$ of highly-specific symptom *loss of smell/taste*, updating $p_{SARS-CoV-2,i}$, the prior probability of SARS-CoV-2 given ARI as determined from virological testing (w = number of ARI reports with symptom; W = total number of ARI reports), using Bayes' theorem (e.g., [18]). This requires estimates of the sensitivity (*sens*) and specificity (*spec*) of the symptom *loss of smell/taste* for SARS-CoV-2 infection; these parameters (0.487 and 0.954, respectively) were taken from a published analysis of UK first wave data collected using the COVID Symptom Study app [13].

$$w_i \sim Binomial(W_i, prev_i)$$
 (6A)

$$\hat{p}_{c,i} = \frac{(sens*p_{c,i})}{(sens*p_{c,i}) + (1 - spec)(1 - p_{c,i})}$$
(6B)

Because this updated prior applies only to individuals who reported loss of smell/taste in week *i*, we define the probability that the ARI report was due to SARS-CoV-2 infection, π_{SARS} $_{-CoV-2,i}$, as a weighted function of the prevalence of loss of smell/taste in that week, and we set $\pi_{influenza,i}$ equal to $p_{influenza,i}$.

$$\pi_{c,i} = (1 - prev_i)p_{c,i} + prev_i(\hat{p}_{c,i}) \tag{7}$$

Implementation

After discarding the first 10,000 iterations as burn-in, 3000 samples from the posterior distribution were taken to allow inference of the model parameters. Three chains were run, and results examined for adequate mixing and convergence (see <u>Supporting Information, S4 Fig</u> for posterior densities for key parameters). We report (and plot) posterior median estimates and define 95% credible intervals (CrIs) using the 2.5% and 97.5% quantiles of the posterior distribution. Sampling was carried out within JAGS [17], accessed via the runjags package [19] for R [20]. JAGS code is provided in the Supporting Information, <u>S1 Code</u>.

To prevent information flow from the observed data on weekly ARI syndrome reports to the posterior probabilities that the ARI report was due to influenza virus or to SARS-CoV-2 infection (i.e., the posterior distributions for the parameters $\pi_{influenza,i}$ and $\pi_{SARS-CoV-2,i}$ should reflect the number of positives and number of tests only), virological testing data are

introduced in a separate JAGS 'data block'. As only one (random) sample from the parameters $p_{c,i}$ are taken per model run (in the JAGS data block), we aggregated the posterior samples from 200 runs of the model, from which the posterior medians and 95% CrIs were extracted.

Given that the Infectieradar weekly survey also queried whether participants had undergone testing (SARS-CoV-2 PCR/antigen test) since their previous survey, we could compare the modelled proportion of ARI patients attributed to SARS-CoV-2 infection to the observed proportion of Infectieradar participants matching the ARI case definition who in the same week (or in the subsequent week, as they may not have tested immediately upon symptom onset) reported a positive test result.

Sensitivity analyses

Sensitivity analysis 1. For application of the approach to other settings, it is of interest to determine the impact of data availability–namely, as mentioned in the Introduction, the total number of swabbed ARI patients from sentinel surveillance undergoing virological testing–on the attribution estimates. This will allow estimation of the minimum weekly number of tested patients required for the obtaining valid attribution estimates from the model. We conducted a sensitivity analysis by successively reducing the total size of the testing dataset through the range (90% through 10% of the original size, in steps of 20%) by randomly removing data points (e.g., 90% of 1074 = 107 removed, 70% of 1074 = 322 removed, etc.), and re-running the model. For each step size, we ran the random removal procedure 20 times, aggregating the posterior samples for the 20 runs, and then calculated the posterior median influenza virus and SARS-CoV-2 attribution estimates and associated 95% credible intervals. We also calculated the root mean-squared error (RMSE), which quantifies the degree of discrepancy to the attribution estimations derived using the full dataset (S1 Table). To give greater weight to larger than smaller discrepancies, we chose RMSE as the loss function.

Sensitivity analysis 2. Prior distribution assumptions were assessed for the probability parameters $p_{c,i}$, by substituting the Jeffreys prior, Beta(0.5,0.5) with the uniform Bayes-Laplace prior, Beta(1,1) (sensitivity analysis 2A), and for the scaling factors δ_c , by substituting the vague half-Normal prior with an alternative, Jeffrey's prior for the Poisson rate parameter (which is proportional to $\delta_c^{-0.5}$) (sensitivity analysis 2B), that also ensured a positive density over the non-negative real numbers.

Results

A total of 11,312 syndrome reports fulfilled the ARI case definition during our 25-week analysis period (per-week mean of 452), out of a total of submitted 229,299 surveys (per-week mean of 9172). The age-group 45–64 years contributed the highest percentage of ARI syndrome reports during weeks 1–25 of 2022 (54%), followed by the 15–44 years age-group (25%).

The proportions of influenza virus and SARS-CoV-2 positive results from virological testing of ARI patients from the sentinel GP network, with fitted cubic smoothing splines, are shown in Fig 1. In comparison with SARS-CoV-2, influenza activity appeared to rise more steeply, peaking approximately 7–8 weeks later. Over the 25-week analysis period, influenza virus and SARS-CoV-2 were detected in 30.6% and 7.8% of samples, respectively. Other viruses were detected as follows: RSV (4.6%), rhinovirus (13.9%), enterovirus (1.2%), other human coronaviruses (9.7%), parainfluenza viruses (7.5%), and human metapneumovirus (1.9%) (NB. double infections could occur). The highest percentage of ARI patients swabbed between 2021/week 40 and 2022 week 20 was for the age-group 15–44 years (approximately 35%), followed by the 45–64 years age-group (approximately 27%) [5].



Fig 1. Weekly observed proportions of influenza-positive and SARS-CoV-2 positive results among sampled ARI patients from sentinel GP surveillance (i.e., number of virus-positive results divided by the number of samples tested per week). Cubic smoothing splines (10 df) are superimposed.

https://doi.org/10.1371/journal.pdig.0000655.g001

The model-estimated number of ARI syndrome reports per week attributable to infection with influenza virus, SARS-CoV-2 infection, or to other causes is depicted in Fig 2. Of the total ARI reports during the defined influenza season (week 8 through week 20), the model





https://doi.org/10.1371/journal.pdig.0000655.g002



Fig 3. Posterior estimates from the attribution model of the weekly proportions per cause (out of total weekly submitted surveys), over the 25-week analysis period.

https://doi.org/10.1371/journal.pdig.0000655.g003

attributed 52.7% (95% CrI: 45.1–58.2%) and 21.3% (95% CrI: 14.6–29.9%) to influenza virus and SARS-CoV-2, respectively. Fig 3 shows the estimated proportions of the total surveys submitted in each week that were attributed to influenza virus, to SARS-CoV-2 and to another cause. Over the whole analysis period, 35.4% (95% CrI: 29.2–40.4%) and 27.0% (95% CrI, 19.3–35.2%) of ARI was estimated to be attributable respectively to influenza virus and to SARS-CoV-2 (but we note the limitations-see Methods and Discussion –in interpreting the attributed percentages given the protocol for selecting patients for swabbing, and differences in healthcare-seeking behaviour between persons who tested SARS-CoV-2 positive using a home test and those who did not). Influenza-attributable ARI reached a peak of 5.0% of all submitted surveys (in week 11), whereas SARS-CoV-2 attributable ARI did not exceed 3.1% (in week 8) within our analysis period (Fig 3).

Posterior estimates of the scaling factors, δ_c , were 707 (95% CrI: 579–806), 884 (95% CrI: 603–1269), and 252 (95% CrI: 209–300) per unit of attribution probability for influenza virus, SARS-CoV-2, and other causes, respectively.

Of those Infectieradar participants with ARI who also reported a linked SARS-CoV-2 PCR/ antigen test, 26.5% had a positive test outcome. Both model-estimated attribution probabilities and the observed proportion of test-positives are plotted over time in <u>S3 Fig</u>, and indicate broadly consistent temporal trends.

Integrating the prevalence of *loss of smell/taste* in the estimation of the probability of SARS-CoV-2 given ARI made only a small difference to the results (27.0% of total ARI reports

over the 25-week analysis period was attributed to SARS-CoV-2 when *loss of smell/taste* was included, compared to 22.6% if excluded), as the prevalence of these highly specific symptoms for SARS-CoV-2 was low throughout the analysis period, reaching their highest levels of 7–8% in week 7 (S2 Fig).

Sensitivity analyses

S1 Table shows the effect of artificially reducing the size of the virological dataset. The uncertainty in the influenza and SARS-CoV-2 attribution estimates for the defined 13-week influenza season tended to increase with decreasing size of the testing dataset. As well, the degree to which attribution proportions differed from those estimated using 100% of the data, as quantified by RMSE, tended to increase as dataset size decreased. Between 50% and 30% of the original size, there was a relatively large increment in RMSE for influenza virus, with a large increment in RMSE at between 70% and 50% for SARS-CoV-2, which suggests that 70% of our full dataset (i.e., mean of 30 tested ARI patients per week) is a practical minimum for attributing respiratory syndrome to influenza or SARS-CoV-2 (S1 Table).

There was minimal influence from the choice of noninformative prior for the proportions of influenza or SARS-CoV-2 positive lab test results, and the alternative prior choice for the scaling factors δ_c also did not substantively alter the attribution estimates (S3 Table).

Discussion

We have presented a statistical modelling approach for the attribution of syndrome selfreports from participatory surveillance to their cause during periods when influenza virus and SARS-CoV-2 are co-circulating, which cannot be adequately distinguished using syndromeonly case definitions. We tested the attribution method using data from Infectieradar and from virological testing of ARI patients from sentinel GP surveillance. Conditional on limitations related to the criteria and the protocol for swabbing ARI patients, the model suggested that a much higher proportion of ARI reports were estimated to be attributed to influenza (53%) than to SARS-CoV-2 (21%) over the 13-week 2022 influenza season in the Netherlands.

Influenza virus activity peaked around week 14–15 (based on the GP sentinel surveillance ILI incidence rate and the test-positive proportion [5]; see Fig 1 in this reference), and there were two COVID-19 peaks that occurred earlier than the influenza peak (according to the weekly number of confirmed SARS-CoV-2 tests at public health service test locations): in week 5 and a smaller peak in week 10 [5]. The time-series of Infectieradar ARI syndrome reports over weeks 1–25 of 2022 therefore reflect a mixture of these two circulating viruses, as well as a sizeable contribution from other (unknown) causal agents (Fig 2). Among those Infectieradar participants whose symptoms matched the ARI case definition who also reported being tested, 26.5% were SARS-CoV-2 positive. This is identical to our overall SARS-CoV-2 attributed proportion (27.0%), and the temporal patterns were roughly consistent (S3 Fig). These similarities, however, should not be interpreted as validation of the model-based attribution estimates given the small proportion of Infectieradar participants who self-selected for, and reported, testing, and the constraints on the sampling of sentinel GP patients and healthcare-seeking behaviour already mentioned (see Methods).

Differences in the estimated scaling factors for influenza virus and SARS-CoV-2 can be cautiously interpreted as between-pathogen differences in the chance that ARI symptoms develop following infection with the particular respiratory pathogen. However, this interpretation must be considered in the context of other possible-but not tested for-causes of the symptoms. Given the ecological design and the constraints on swabbing ARI patients in our data, one cannot definitively say on the basis of a larger scaling factor estimated for influenza that influenza virus infection leads to a higher chance of ARI symptoms compared with SARS-CoV-2.

Advantages of the present approach, which is similar to ecological regression methods employed for attribution of respiratory-associated mortality or hospital admissions over multiple seasons [3,7,21], is that it is relatively simple to implement and requires only virological testing data from a comparable population. Sensitivity analysis 1 has suggested a lower limit for the dataset size (i.e., a mean of 30 ARI patients tested per week) needed to attribute respiratory syndrome occurrence to the causal agent. Unlike additive regression approaches, data from multiple seasons are not required for robust estimates. Furthermore, the Bayesian modelling framework propagates uncertainty from all data sources to the attribution estimates, the prevalence of distinguishing symptoms is straightforward to incorporate, and the approach allows further evidence to be integrated if available; for instance, a second source of weekly SARS-CoV-2 testing data could easily be included.

The main limitation is the ecological approach to estimation, and the consequent key assumption that the population sampled for virological testing is comparable–in susceptibility to infection, symptom severity, and in demographic characteristics such as age–to the respondents reporting the syndrome of interest. Implicit in the use of sentinel GP data is the assumption that health-care seeking behaviour is not conditional on the cause; for instance, if one strongly suspects SARS-CoV-2 infection and goes to a government COVID-19 testing facility rather than to a GP, then the percentage of ARI attributed to SARS-CoV-2 will be under-estimated. Because patients were advised to do a rapid antigen test before going to the GP, and persons testing positive were neither recorded as ARI patients nor swabbed, our virological testing dataset is biased against detecting SARS-CoV-2. Future work will address the validity of the attribution model, as in the following (2022/2023) season a random sample of Infectieradar participants were mailed a rapid antigen test which they were asked to use if they had respiratory symptoms, and–if the result was SARS-CoV-2 positive–to send in the swab to IDS for testing for other viruses including as influenza. This will provide data on SARS-CoV-2 and influenza infection within a single study population.

The tested population from the sentinel GP network consisted of a comparable number of ILI patients and non-ILI ARI patients [5], which may not provide an ideal match to the Infectieradar ARI reports. In addition, because the absolute numbers of syndrome reports attributed to each cause will vary depending on the syndrome case definition (e.g., greater weekly numbers of influenza- and SARS-CoV-2 attributed syndrome reports would be estimated when based on COVID-19-like illness instead of ARI, or on the more specific ILI case definition; S1 Fig), the proportion of syndrome reports attributed to each cause is a more appropriate outcome measure. We have made some simplifications for the purpose of illustrating the method. Namely, the sensitivity and specificity of loss of smell/taste are likely variant-specific; using the first wave (ancestral variant) estimates for the current omicron-era data will introduce some inaccuracies. Finally, the time-series of both ARI syndrome reports and virological testing are assumed to align; however, the onset of ARI (in both Infectieradar and sentinel GP patients) may occur a few days before being tested.

In summary, the proposed method can assist in attributing respiratory syndrome reports from participatory surveillance to either influenza virus or SARS-CoV-2 infection in periods when both viruses are circulating. The approach is particularly interesting because it can be easily exported to other countries with implemented participatory surveillance systems. The estimated lower bound in the weekly number of swabs submitted for virological testing provided by sensitivity analysis 1 supports the feasibility of the present approach in other countries where participatory surveillance data are available but routine virological testing is less well-established than in the Netherlands.

Supporting information

S1 Table. Result of sensitivity analysis 1, in which the size of the virological testing dataset is varied.

(DOCX)

S2 Table. Data from Infectieradar participatory surveillance. (DOCX)

S3 Table. Result of sensitivity analysis 2, in which alternative prior distributions for two parameters are tested.

(DOCX)

S1 Fig. Weekly prevalence of three respiratory syndromes in weeks 1–25 of 2022. (DOCX)

S2 Fig. Prevalence of symptoms highly specific for SARS-CoV-2 infection. (DOCX)

S3 Fig. Comparison of the proportion of weekly Infectieradar ARI reports attributed to SARS-CoV-2 with the proportion of weekly ARI reports with a linked positive test result. (DOCX)

S4 Fig. Trace plots and posterior densities for seven key model parameters. (DOCX)

S1 Code. JAGS code. (DOCX)

Acknowledgments

Infectieradar team group author: Kirsten Bulsink, Fleur Petit, Thomas Dalhuisen, Steven Nijman, Liz Jenniskens, Tara Smit, Danytza Berry, Denise Wong, Yolanda van Weert

We thank the participating general practitioners of the sentinel network of the Netherlands Institute for Health Services Research who provided swabs from ARI patients, as well as lab personnel at RIVM who carried out virological analyses.

Author Contributions

Conceptualization: Scott A. McDonald, Daniela Paolotti.

Data curation: Marit de Lange.

Formal analysis: Scott A. McDonald.

Investigation: Albert Jan van Hoek, Mariette Hooiveld, Adam Meijer, Arianne van Gageldonk-Lafeber, Jacco Wallinga.

Methodology: Scott A. McDonald, Daniela Paolotti.

Project administration: Albert Jan van Hoek.

Resources: Adam Meijer.

Writing - original draft: Scott A. McDonald.

Writing – review & editing: Scott A. McDonald, Albert Jan van Hoek, Daniela Paolotti, Mariette Hooiveld, Adam Meijer, Marit de Lange, Arianne van Gageldonk-Lafeber, Jacco Wallinga.

References

- van Noort SP, Codeco CT, Koppeschaar CE, Van Ranst M, Paolotti D, Gomes MGM. Ten-year performance of Influenzanet: ILI time series, risks, vaccine effects, and care-seeking behaviour. Epidemics 2015; 13:28–36. https://doi.org/10.1016/j.epidem.2015.05.001 PMID: 26616039
- Paolotti D, Carnahan A, Colizza V, Eames K, Edmunds J, Gomes G, et al. Web-based participatory surveillance of infectious diseases: the Influenzanet participatory surveillance experience. Clin Microbiol Infect. 2014; 20:17–21. https://doi.org/10.1111/1469-0691.12477 PMID: 24350723
- Friesema IHM, Koppeschaar CE, Donker GA, Dijkstra F, Van Noort SP, Smallenburg R, et al. Internetbased monitoring of influenza-like illness in the general population: experience of five influenza seasons in The Netherlands. Vaccine 2009; 27:6353–6357. <u>https://doi.org/10.1016/j.vaccine.2009.05.042</u> PMID: 19840672
- Vandendijck Y, Faes C, Hens N. Eight years of the Great Influenza Survey to monitor influenza-like illness in Flanders. PLoS One 2013; 8:e64156. <u>https://doi.org/10.1371/journal.pone.0064156</u> PMID: 23691162
- Reukers DFM, van Asten L, Brandsema PS, Dijkstra F, Hendriksen JMT, Hooiveld M, et al. Surveillance of influenza and other respiratory infections in the Netherlands: Winter 2021/2022. Bilthoven NL: RIVM; 2022.
- RIVM. Open Data Repository. National COVID-19 cases; 2022. URL: https://data.rivm.nl/covid-19/ COVID-19_casus_landelijk.html.
- Van Asten L, Van Den Wijngaard C, Van Pelt W, Van De Kassteele J, Meijer A, Van Der Hoek W, et al. Mortality attributable to 9 common infections: significant effect of influenza A, respiratory syncytial virus, influenza B, norovirus, and parainfluenza in elderly persons. J Infect Dis 2012; 206:628–639. https://doi. org/10.1093/infdis/jis415 PMID: 22723641
- Green HK, Andrews N, Fleming D, Zambon M, Pebody R. Mortality attributable to influenza in England and Wales prior to, during and after the 2009 pandemic. PloS One 2013; 8:e79360. <u>https://doi.org/10. 1371/journal.pone.0079360</u> PMID: 24348993
- European Centre for Disease Control (ECDC). Case definition for coronavirus disease 2019 (COVID-19), as of 3 December 2020. ECDC, 2020. https://www.ecdc.europa.eu/en/covid-19/surveillance/casedefinition (Accessed 10 August 2022).
- European Union (EU). Commission implementing decision (EU) 2018/945 of 22 June 2018 on the communicable diseases and related special health issues to be covered by epidemiological surveillance as well as relevant case definitions. EUR-Lex, L170-1. EU, 2018. Available at: http://data.europa.eu/eli/ dec_impl/2018/945/oj.https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:32018D0945
- Guerrisi C, Turbelin C, Blanchon T, Hanslik T, Bonmarin I, Levy-Bruhl D, et al. Participatory syndromic surveillance of influenza in Europe. J Infect Dis. 2016; 214:S386–S392. <u>https://doi.org/10.1093/infdis/jiw280 PMID: 28830105</u>
- McDonald SA, Van Den Wijngaard CC, Wielders CC, Friesema IH, Soetens L, Paolotti D, et al. Risk factors associated with the incidence of self-reported COVID-19-like illness: data from a web-based syndromic surveillance system in the Netherlands. Epidemiol Infect. 2021;149. <u>https://doi.org/10.1017/S0950268821001187 PMID: 34006340</u>
- Antonelli M, Capdevila J, Chaudhari A, Granerod J, Canas LS, Graham MS, et al. Optimal symptom combinations to aid COVID-19 case identification: Analysis from a community-based, prospective, observational cohort. J Infect. 2021; 82:384–390. <u>https://doi.org/10.1016/j.jinf.2021.02.015</u> PMID: 33592254
- Zayet S, Lepiller Q, Zahra H, Royer PY, Toko L, Gendrin V, et al. Clinical features of COVID-19 and influenza: a comparative study on Nord Franche-Comte cluster. Microbes Infect. 2020; 22:481–488. https://doi.org/10.1016/j.micinf.2020.05.016 PMID: 32561409
- 15. Donker GA. Nivel Primary Care Database—Sentinel Practices 2014. Utrecht: Nivel; 2016.
- Teirlinck AC, De Gier B, Meijer A, Donker G, De Lange M, Koppeschaar C, et al. The incidence of symptomatic infection with influenza virus in the Netherlands 2011/2012 through 2016/2017, estimated using Bayesian evidence synthesis. Epidemiol Infect. 2019;147.
- Plummer M. JAGS: a program for analysis of Bayesian graphical models using Gibbs sampling. In Proceedings of the 3rd International Workshop on Distributed Statistical Computing; 2003. Vol. 124, No. 125, p.10.
- Sox H, Stern S, Owens D, Abrams HL Chapter 2, The use of diagnostic tests: a probabilistic approach. Assessment of diagnostic technology in health care: Rationale, methods, problems, and directions: Monograph of the Council on Health Care Technology. Washington (DC): National Academies Press (US); 1989.

- 19. Denwood MJ. Runjags: an R package providing interface utilities, model templates, parallel computing methods and additional distributions for MCMC models in JAGS. J Stat Software 2016; 71:1–25.
- 20. R Core Team. R: A Language and Environment for Statistical Computing. Vienna, Austria: R Foundation for Statistical Computing; 2022.
- Cromer D, Van Hoek AJ, Jit M, Edmunds WJ, Fleming D, Miller E. The burden of influenza in England by age and clinical risk group: a statistical analysis to inform vaccine policy. J Infect. 2014; 68:363–371. https://doi.org/10.1016/j.jinf.2013.11.013 PMID: 24291062